\begin{table}%tA.2
\caption{\label{tab10} SN~Ib, Ic and II  template spectra in ${\cal SN}$-Fit database.}
%\centerline
{ \begin{tabular}{cccc}
\hline\hline
 SN & Type  & Number of & Reference\\ 
Name && spectra &\\
\hline
\object{SN~1983N} & Ib  & 2       & \citet{Harkness87}\\
\object{SN~1984L} & Ib & 5        & \citet{Harkness87}\\
\object{SN~1987A} & II\_pec & 4   & \citet{Pun95}\\    
\object{SN~1987M} & Ic & 1        & \citet{Filippenko90}\\
\object{SN~1990B} & Ic & 4        & \citet{Matheson01}\\
\object{SN~1990U} & Ic & 3       & \citet{Matheson01}\\
\object{SN~1993J} & IIb & 6       & \citet{Matheson00}\\
\object{SN~1994I} & Ic & 7        & \citet{Filippenko95}\\
\object{SN~1997ef}&  Ibc\_pec & 3 & \citet{Matheson01}\\
\object{SN~1998bw}& Ic & 7        & \citet{Patat01}\\  
\object{SN~1998dt}& Ib & 2        & \citet{Matheson01}\\
\object{SN~1999di}& Ib & 2        & \citet{Matheson01}\\
\object{SN~1999dn}& Ib & 3        & \citet{Matheson01}\\
\object{SN~1999em}& IIp & 11      & \citet{Hamuy01}\\  
\object{SN~1999ex}& Ibc & 3       &\citet{Hamuy02}\\  
\object{SN~2000h} & Ib & 1        & \citet{Branch02}\\ 
\hline
Total Ib/Ic = 43   &Total II~=~21 &  Total Ib/Ic/II = 64. \\
\end{tabular}}
\par
\end{table}